Author’s response to reviews

Title: Content analysis of press coverage during pandemic influenza H1N1 in Germany 2009/2010

Authors:

Sabine Husemann (Sabine-Husemann@gmx.de)
Florian Fischer (f.fischer@uni-bielefeld.de)

Version: 2 Date: 11 February 2015

Author’s response to reviews:

Dear sirs or madams,

we have added the page numbers and expanded the line numbering also to the abstract.

We changed the wording of "Introduction" into "Background".

We analysed newspaper articles in a content analysis. Therefore, no ethical or consent statement is needed.

A list of abbreviations (RKI and WHO) was added.